# Geometric View on Integrated Cascaded Channel of IRS-Aided Communications

Yunli Li<sup>10</sup> and Young Jin Chun<sup>10</sup>, Member, IEEE

Abstract— The hybrid intelligent reflecting surface (IRS) architecture is a novel technology that leverages the advantages of both passive and active IRS; the passive IRS offers a large aperture, while the active IRS provides additional power amplification. Prior studies have shown that the optimal performance of IRS-assisted wireless networks is achieved when the passive IRS is deployed near the transceivers and the active IRS is closer to the receiver with decreasing amplification power, assuming that the IRS is located between transceivers and that the height difference between IRS and transceivers is much smaller than the link distance between transceivers. However, most of the prior works on IRS blindly adopted this assumption in the IRS association policy, which essentially becomes a partial selection strategy that offers analytical simplicity at the cost of suboptimal performance. This limitation motivated us to find the globally optimal deployment strategy for all types of IRS. To this end, we first employ the geometric models for integrated path loss distance (known as Cassini oval and Ellipse for productand sum-distance path loss laws, respectively) and use them to determine the optimal locations of the hybrid IRS. Then, we design a novel opportunistic association policy for hybrid IRS based on the integrated path loss model. Furthermore, we validate our proposed methods through simulations and show that they significantly outperform the conventional nearest association policy, especially for hybrid and active IRS.

*Index Terms*—Hybrid IRS, opportunistic association policy, geometric models, integrated path loss distance, cascaded channel.

## I. INTRODUCTION

THE Intelligent Reflecting Surface (IRS), also known as Reconfigurable Intelligent Surface (RIS) and Large-Scale Intelligent Surface (LIS), is considered as a critical technology capable of dynamically altering signal paths, extending coverage, and simultaneously enhancing network capacity for future wireless networks [1], [2], [3], [4], [5]. There are three types of IRS architecture: passive IRS, active IRS, and hybrid IRS.

Manuscript received 24 September 2023; revised 24 January 2024; accepted 18 March 2024. Date of publication 9 April 2024; date of current version 12 September 2024. This work was supported in part by the Early Career Scheme (ECS) under Project 21205021 and the General Research Fund (GRF) under Project 11211122, both established under the University Grant Committee (UGC) of the Hong Kong Special Administrative Region, China; in part by the City University of Hong Kong (CityU) under Project 7020083, and Project 7006090. The associate editor coordinating the review of this article and approving it for publication was A. Guerra. (*Corresponding author: Young Jin Chun.*)

The authors are with the Department of Electrical Engineering, City University of Hong Kong, Hong Kong, China (e-mail: yunlili2-c@my.cityu.edu.hk; yjchun@cityu.edu.hk).

Color versions of one or more figures in this article are available at https://doi.org/10.1109/TWC.2024.3384185.

Digital Object Identifier 10.1109/TWC.2024.3384185

The passive IRS consists of a phase shift controller and a large surface equipped with numerous reflecting elements. The passive reflecting elements merely reflect signals without additional active processing, which enables high spectraland energy-efficient communication at a low cost [1], [6], [7]. Despite the performance gain provided by passive IRS, the enhancement can be significantly impeded by the severe two-fold path loss. Specifically, the benefit of passive IRS is limited when the direct link between transceivers is stronger than the IRS-aided links [8], [9].

Active IRS can overcome the severe path loss by simultaneously reflecting and amplifying the received signals [10], [11], [12], [13], [14], [15]. The integration of amplifiers into the active reflecting elements can be achieved through various existing active components, such as current-inverting converters [16], asymmetric current mirrors [17] or integrated circuits [18]. Nevertheless, unlike passive IRS that operates in a noise-free mode, the aperture of active IRS is constrained by the amplification power and thermal noise generated by active parts.

To combine the benefits of passive and active IRS, a novel hybrid IRS architecture is proposed in [19], which contains two sub-surfaces: one of the sub-surfaces is composed of passive reflecting elements, while the other sub-surface consists of active reflecting elements and additional operation power is required.<sup>1</sup> The hybrid IRS leverages the advantages of both passive and active IRS in a complementary manner. Specifically, the passive IRS offers asymptotic beamforming gain with a squared power scaling order of  $O(N^2)$ , while the active IRS provides an extra power amplification gain but with a linear power scaling order of O(N) for beamforming due to thermal noise introduced by the active parts, where N is the number of reflecting elements [21].

However, implementing the innovative hybrid architecture presents a difficult issue due to disparate location preferences for passive and active sub-surfaces. Specifically, the passive sub-surface is inclined towards being situated in close proximity to the transceivers, while the active sub-surface necessitates placement at a certain distance away from the transmitter to ensure an acceptable amplification factor [8], [21]. Furthermore, active sub-surface is supposed to be deployed closer to the receiver with decreasing amplification power [21]. Consequently, finding the optimal deployment strategy is one of the

<sup>1</sup>The definition of hybrid IRS is different from [20], where the active reflecting element are connected an RF chain to support channel estimation and without amplification.

© 2024 The Authors. This work is licensed under a Creative Commons Attribution 4.0 License. For more information, see https://creativecommons.org/licenses/by/4.0/ most essential design challenges for the hybrid IRS-assisted wireless networks to achieve the full potential of both passive and active sub-surfaces.

The channel properties are crucial for analyzing the endto-end performance of IRS-aided communications. As summarized in [22], there are numerous small-scale multipath fading models, including the phase shift model, transmissionmode model, physical-based channel model, beamspace channel model, geometry-based stochastic model, keyhole channel model, and machine learning-based channel model. Rician fading model, especially, is popular in IRS-assisted communication due to its line-of-sight assumption [23], [24]. In [25], the authors proposed a beamspace model based on extended Saleh-Valenzuela (SV) channel, categorizing the scatters and paths of multipath components. Mixture Gamma distribution-based, generalized fading model was introduced in [9] to ease the analysis of IRS-aided networks.

To the best of our knowledge, the geometric modeling of the integrated path loss distance remains an open problem, although many studies have modeled small-scale fading in the cascaded channel. To unlock the full potential of hybrid IRS, it is crucial to comprehend the geometric arrangement of node locations for optimal IRS placement. Furthermore, we need to develop an opportunistic association policy for IRS, which associates IRS based on the best end-to-end performance [26].

#### A. Geometric Model

The geometric model offers a realistic representation of wireless networks and provides a spatial understanding of the given wireless network. The location of nodes affects signal strength, interference, and ultimately the performance of the network. The geometric model can accurately portray the physical distances and relative positions of nodes, allowing for better predictability of system behavior [27].

Comprehending the geometric nature of a network enables the anticipation of signal propagation, and identification of interference patterns and potential bottlenecks. This understanding proves invaluable in optimizing the deployment and node density to enhance overall network performance [27]. For example, in machine-to-machine (M2M) communications, where the fixed locations of user equipment (UE) allow for the strategic design of IRS deployment based on defined objective functions, such as maximizing IRS utility or minimizing interference among UEs. The geometric model further plays a pivotal role in informing decisions regarding the integration of a new node. For instance, once the system's deployment is established, UEs can access a look-up table of potential average performance for each IRS from the base station (BS) upon joining the cell and reporting their locations. Subsequently, UE can judiciously select the serving IRS based on the information provided in the table.

Furthermore, the geometric model exhibits adaptability across diverse wireless scenarios, encompassing both mobile and static networks, while seamlessly accommodating variations in network size and topology. Notably, in IRS-aided networks, geometric model is of critical importance in enhancing the analytical framework by precisely capturing the statistics of the integrated links based on practical opportunistic association policies, as opposed to relying on partial selection over the nearest association. Moreover, the geometric model serves as a robust foundation for stochastic geometrybased system-level analysis. This is attributed to their ability to model the distribution of connection distances among nodes based on their inherent geometric relationships [28], [29], [30].

The ellipse model, extensively explored for sum-distance, has gained much attention in localization problems. However, to the best of the authors' knowledge, this is the first exploration of the Cassini oval model for product-distance in the realm of wireless communications.

#### B. Association Policy: Nearest and Opportunistic

For IRS-aided communication, accurate statistical modeling of the integrated cascaded channel remains an open problem due to the lack of exploration in the integrated path loss distance. Most existing literature relies on the nearest association policy, which redirects the focus to individual link by trading accuracy for tractability [6], [7], [9].

The nearest association policy entails a partial selection mechanism for passive IRS grounded by a simple setup with the product-distance path loss law, where the location of the IRS is confined to the segment between the BS and UE with limited height [1], [31]. Additionally, this association relies on a widely adopted common assumption for analytical simplification, where the link distance between the associated passive IRS and the serving BS of the typical UE, referred to as the BS $\rightarrow$ IRS link, equals the link distance between the BS and UE, denoted as BS $\rightarrow$ UE link. This simplification directs the analysis toward the link distance between the IRS and UE, denoted as IRS $\rightarrow$ UE link, rather than considering the integrated cascaded BS $\rightarrow$ IRS $\rightarrow$ UE channel.

However, this assumption proves impractical for several reasons. First, the approximated nearest association is effective solely for the product-distance path loss law. Yet, diverse propagation scenarios necessitate distinct path loss laws, as corroborated by [32]. For example, the product-distance path loss law is appropriate for far-field and near-field beamforming cases, whereas the sum-distance path loss law is suitable for near-field broadcasting cases. Regrettably, the nearest association policy is ill-suited for the sum-distance path loss law. Second, in IRS-aided wireless networks, the channel condition of the integrated cascaded channel, i.e., BS-JIRS-JUE, directly governs end-to-end communication quality, instead of the individual links, i.e., BS -> IRS or IRS $\rightarrow$ UE. Specifically, even if one of the single links attains sufficient quality, poor performance may persist if the other link is obstructed.

Moreover, the nearest association policy fails to realize full potential of the active and hybrid IRS. As illustrated in [21], the active IRS should be deployed closer to the receiver with decreasing amplification power to maximize the achievable rate. In addition, for hybrid IRS, the deployment preferences of the two sub-surfaces, namely passive and active, along the segment between the BS and UE are different. As such, within this constrained deployment framework along the segment between BS and UE, a performance balance arises between these two sub-surfaces regarding to IRS locations. This conflict deployment preference between the two sub-surfaces has motivated us to investigate the optimal locations and opportunistic association of hybrid IRS across the entire area, which requires comprehensive understanding of the integrated channel.

The integrated cascaded channel imposes elevated demands on the accurate modeling of communication, necessitating a delicate equilibrium between analytical complexity and accuracy. Our exploration leverages the geometric properties of the integrated path loss distance within the IRS-aided cascaded channel. While previous work, as outlined in [33], has delved into the optimal selection of passive IRS in a single-cell network using stochastic geometry tools, further research into the network's geometric nature is imperative for studying the optimal locations of the hybrid IRS. Specifically, we need to investigate the cascaded channel's end-to-end performance from a geometric perspective, which is crucial for developing an opportunistic association policy and a deployment strategy tailored for hybrid IRS.

## C. Contributions

Driven by the aforementioned concerns, our central objective is to address the *geometric modeling* intricacies inherent in the equal-gain integrated path loss distance within the IRS-assisted cascaded channel. Subsequently, we formulate an optimization problem to ascertain the optimal locations for the hybrid IRS and put forth an *opportunistic association policy* tailored for hybrid IRS. The main contributions of this paper may be summarized as follows:

- We examine two geometric models of the integrated path loss distance for an IRS-aided cascaded channel, referred to as Cassini oval and Ellipse for productand sum-distance path loss laws, respectively. Building upon the geometric models, we explore the statistical characteristics of the integrated path loss distance, such as equal-gain trajectory, cumulative distribution function (CDF) and probability density function (PDF).
- 2) Based on the geometric nature of node locations, we demonstrate that the optimal locations for the hybrid IRS can be determined by the integrated path loss distance of the cascaded channel (BS→IRS→UE), system parameters, and the link distance between the BS and UE. Moreover, the outcomes for both passive and active IRS can be obtained as special instances of the hybrid IRS.
- 3) We investigate potential implementations of the proposed geometric models in IRS-aided networks and compare the performance of opportunistic and nearest association policies for IRS. It is noted that the nearest association policy is ineffective for hybrid/active IRS, while the opportunistic association policy results in significant performance improvement across all types of IRS.

## D. Organizations

The remaining paper is organized as follows. Section II introduces the system model, Section III presents the geo-



Fig. 1. Downlink of a hybrid IRS-aided network. The configuration of the IRS effective zone for transceiver pairs undergoes variations based on the ratio of the integrated path loss distance of the cascaded channel to the link distance between the BS and UE. In accordance with the IRS effective zone, the IRSs are categorized into three types: Type-I, the IRSs located within the IRS effective zone for a single transceiver pair; Type-II, the IRSs positioned within the intersection of the IRS effective zones for more than one transceiver pairs; Type-III, the IRSs situated outside the IRS effective zones.

metric models and derives statistical characteristics for the equal-gain integrated path loss distance of the cascaded channel. Section IV displays the channel statistics in IRS-aided communication, establishes the hybrid IRS's optimal locations and proposes an opportunistic association policy. Section V analyzes special cases to evaluate our proposed geometric model. Section VI provides Monte-Carlo simulations to verify our analysis and Section VII concludes the paper.

#### II. SYSTEM MODEL

We consider a hybrid IRS-aided network, as illustrated in Fig. 1, where both BSs and UEs are equipped with a single antenna and orthogonal multiple access is employed within each cell. Without loss of generality, we assume that each BS has an infinitely backlogged queue, i.e., the BS always has data to transmit to each user, [26], [34]. Each hybrid IRS has N reflecting elements and contains two sub-surfaces. One of the sub-surfaces contains  $N_{\text{pas}}$  passive reflecting elements along with a phase-shift controller, while the other sub-surface contains  $N_{\text{act}}$  active reflecting elements, a phase-shift controller, and an additional amplify controller, as shown in Fig. 2 [19]. We assume that J hybrid IRSs are uniformly distributed throughout the whole area, and we denote the set of IRSs as  $\mathcal{J} \triangleq \{\text{IRS}_1, \cdots, \text{IRS}_J\}$ .

The reflection matrix serves as a crucial distinguishing factor between passive and active sub-surfaces. Specifically, the reflection matrix of the passive sub-surface is presented as  $\Psi_{\text{pas}} \triangleq \text{diag}(e^{j\phi_1^{\text{pas}}}, \cdots, e^{j\phi_{N_{\text{pas}}}^{\text{pas}}}) \in \mathbb{C}^{N_{\text{pas}} \times N_{\text{pas}}}$ , where  $\phi_{n_{\text{pas}}}^{\text{pas}}$  represents the phase shift of the  $n_{\text{pas}}$ -th passive reflecting element with  $n_{\text{pas}} \in \mathcal{N}_{\text{pas}} \triangleq \{1, \cdots, N_{\text{pas}}\}$ , and the reflection amplitude of the passive reflecting elements are set as one. In contrast, the reflection matrix of the active sub-surface is denoted as  $\Psi_{\text{act}}^{(\eta)} \triangleq \text{diag}(\eta_1 e^{j\phi_1^{\text{act}}}, \cdots, \eta_{N_{\text{act}}} e^{j\phi_{N_{\text{act}}}}) \in \mathbb{C}^{N_{\text{act}} \times N_{\text{act}}}$ , where  $\phi_{n_{\text{act}}}^{\text{act}}$  represents the phase shift of the  $n_{\text{act}}$ -th active reflecting element with  $n_{\text{act}} \in \mathcal{N}_{\text{act}} \triangleq \{1, \cdots, N_{\text{act}}\}$ ,  $\eta_{n_{\text{act}}}$  denotes the reflection amplitude of the  $n_{\text{act}}$ -th active reflecting element. Without loss of generality,

for the associated IRS, we assume a common amplification factor across all active reflecting elements,<sup>2</sup> i.e.,  $\eta_{n_{act}}^2 \triangleq \frac{P_F}{N_{act}(P_T \mathbb{E}[G_{BI}]\epsilon d_{BI}^{-\alpha} + \delta_F^2)}, \forall n_{act}$  [21], [35], [36], where  $P_F$  is the amplification power of the active reflecting elements,  $\mathbb{E}[G_{BI}]$  is the average small-scale channel gain of BS $\rightarrow$ IRS link,  $\delta_F^2$  is the thermal noise power generated by the active reflecting elements,  $P_T$  is the transmit power of the BS,  $\epsilon$  represents the reference channel power gain at a distance of 1 meter (m), and  $\alpha$  is the path loss exponent ( $\alpha > 2$ ) [21]. Then, the reflection matrix of the active sub-surface can be further expressed as  $\Psi_{act}^{(\eta)} \triangleq \eta \Psi_{act}$ , where  $\Psi_{act} = \text{diag}(e^{j\phi_1^{act}}, \cdots, e^{j\phi_{N_{act}}^{act}}) \in \mathbb{C}^{N_{act} \times N_{act}}$ .

We examine the cascaded channel between a typical UE and its serving BS via the associated IRS. We present an example case in which the signal reflected by all reflecting elements on the same IRS experiences identical fading and path loss. It is noted that the quality of communication for the cascaded channel is determined by the path loss and the small-scale fading of the integrated cascaded channel (BS $\rightarrow$ IRS $\rightarrow$ UE) rather than those of each individual link. In other words, even if one link of the cascaded channel is lossless, severe blockage of the other link may still render the entire cascaded channel non-functional. Therefore, this work emphasizes the integrated channel condition of the cascaded BS $\rightarrow$ IRS $\rightarrow$ UE channel instead of focusing on individual link conditions, i.e., BS $\rightarrow$ IRS link or IRS $\rightarrow$ UE link. To facilitate subsequent analysis, we define an integrated path loss distance.

Definition 1 (Integrated Path Loss Distance): We define the integrated path loss distance of the IRS-assisted cascaded channel (BS $\rightarrow$ IRS $\rightarrow$ UE) as  $d_{\rm BIU} = d_{\rm BI} \circ d_{\rm IU}$ , where  $\circ$  depends on the adopted path loss law,  $d_{\rm BI}$  and  $d_{\rm IU}$ are the link distances of BS $\rightarrow$ IRS and IRS $\rightarrow$ UE links, respectively. If the product-distance path loss law is assumed, for near-field beamforming or far-field communications, the integrated path loss distance is defined as  $d_{\rm BIU,P} = d_{\rm BI} \times d_{\rm IU}$ ; If the sum-distance path loss law is adopted, for near-field broadcasting, the integrated path loss distance is defined as  $d_{\rm BIU,S} = d_{\rm BI} + d_{\rm IU}$  [1], [32].

The small-scale fading of the cascaded BS $\rightarrow$ IRS $\rightarrow$ UE channel over passive and active sub-surfaces, denoted by  $\mathbf{g}_{\mathrm{BIU}}^{\mathrm{pas}}$ , and  $\mathbf{g}_{\mathrm{BIU}}^{\mathrm{act}}$ , respectively, are modeled as

$$\mathbf{g}_{\mathrm{BIU}}^{\mathrm{pas}} = (\mathbf{g}_{\mathrm{IU}}^{\mathrm{pas}})^H \boldsymbol{\Psi}_{\mathrm{pas}} \mathbf{g}_{\mathrm{BI}}^{\mathrm{pas}}, \quad \mathbf{g}_{\mathrm{BIU}}^{\mathrm{act}} = (\mathbf{g}_{\mathrm{IU}}^{\mathrm{act}})^H \boldsymbol{\Psi}_{\mathrm{act}} \mathbf{g}_{\mathrm{BI}}^{\mathrm{act}}, \quad (1)$$

where  $\mathbf{g}_{\text{BI}}^{\text{pas}} \in \mathbb{C}^{N_{\text{pas}} \times 1}$ ,  $(\mathbf{g}_{\text{IU}}^{\text{pas}})^H \in \mathbb{C}^{1 \times N_{\text{pas}}}$ ,  $\mathbf{g}_{\text{BI}}^{\text{act}} \in \mathbb{C}^{N_{\text{act}} \times 1}$ ,  $(\mathbf{g}_{\text{IU}}^{\text{act}})^H \in \mathbb{C}^{1 \times N_{\text{act}}}$ , and we adopt parameter set (l, j) to denote the link type and the mode of IRS sub-surface;  $l \in \{\text{IU}, \text{BI}\}$  with l = 'IU' representing the IRS $\rightarrow$ UE link and l = 'BI' indicating the BS $\rightarrow$ IRS link. Similarly,  $j \in \{\text{pas, act}\}$  with j = 'pas' denoting the passive IRS sub-surface and j = 'act' indicating the active IRS sub-surface. In (1),  $\mathbf{g}_l^j$  represents small-scale fading across the link l of the employed IRS mode j, where  $n_j \in \mathcal{N}_j \triangleq \{1, \dots, N_j\}, l \in \{\text{IU, BI}\}$ , and  $j \in \{\text{pas, act}\}$ .

Phase-shift Controller Phase-shift and amplify controller

Ņ

N<sub>act</sub>

**N**<sub>pas</sub>

Fig. 2. Hybrid IRS: One sub-surface contains  $N_{\text{pas}}$  passive reflecting elements along with a phase-shift controller, while the other contains  $N_{\text{act}}$  active reflecting elements, a phase-shift controller, and an amplify controller [19].

Furthermore, we consider generalized small-scale fading for each individual link to enable the versatility and flexibility of the results. It was displayed in [37] that most of the existing fading distributions, i.e., Rayleigh, Rician, one-sided Gaussian, Nakagami-m,  $\kappa - \mu$ , and  $\kappa - \mu$  shadowed fading, can be modeled as mixture Gamma distributions with high accuracy and tractability. To this end, we first introduce the definition of mixture Gamma distribution, and the properties of mixture Gamma distributed channels are summarized below [38].

Definition 2 (The Mixture Gamma Distribution): A random variable G that follows a mixture Gamma distribution is denoted as  $G \sim \mathcal{MG}(\varepsilon_i, \beta_i, \xi_i)$ , and its PDF, CDF and moments are given by

$$f_{\rm G}(x) = \sum_{i=1}^{I} \varepsilon_i x^{\beta_i - 1} e^{-\xi_i x}, \qquad (2a)$$

$$F_{\rm G}(x) = \sum_{i=1}^{I} \varepsilon_i \xi_i^{-\beta_i} \gamma(\beta_i, \xi_i x), \qquad (2b)$$

$$\mathbb{E}_{\mathbf{G}}\left[x^{r}\right] = \sum_{i=1}^{I} \varepsilon_{i} \frac{\Gamma(\beta_{i}+r)}{\xi_{i}^{\beta_{i}+r}},\tag{2c}$$

where  $\{\varepsilon_i, \beta_i, \xi_i\}$  are the parameters for each Gamma component,  $\gamma(\cdot, \cdot)$  is the lower incomplete Gamma function.<sup>3</sup>

Moreover, [9] has proved that the cascaded small-scale channel gain can be modeled as a mixture Gamma distribution with high accuracy and tractability.

## III. GEOMETRIC MODELS FOR THE INTEGRATED PATH LOSS DISTANCE

In this section, we study geometric models for the integrated path loss distance under product- and sum-distance path loss laws, as defined in Definition 1. We derive the CDF and PDF of the integrated path loss distance of the cascaded channel based on the proposed geometric models. Herein, we consider an end-to-end communication scenario, where

Passive IRS element

Active IRS element

<sup>&</sup>lt;sup>2</sup>As shown in [19], all active reflecting elements are supposed to adopt identical amplification factor as they suffer the same path loss. Nonetheless, this may not suitable for multi-UE scenario as the beamforming design should balance the tradeoff among numerous UEs.

<sup>&</sup>lt;sup>3</sup>The number of Gamma components, *I*, is usually set as 20 to guarantee sufficient accuracy, i.e., approximation error less than  $10^{-6}$  [9], [37].

the typical UE and its serving BS are symmetrically located along the x-axis at Cartesian coordinates (c, 0) and (-c, 0), respectively. We describe the geometric models for two path loss laws below.

## A. Product-Distance Path Loss Law: Cassini Oval

For product-distance path loss law, the integrated path loss distance of the cascaded channel is defined as the product of individual link distance, i.e.,  $d_{\rm BIU,P} = d_{\rm BI} \times d_{\rm IU}$ , which has been validated in [32] to be suitable for both near-field beamforming and far-field communication scenarios.

Theorem 1 (Equal Product-Distance Trajectory (Cassini Oval)): For the product-distance path loss law, the trajectory of IRSs with the equal integrated path loss distance of the cascaded BS $\rightarrow$ IRS $\rightarrow$ UE channel,  $d_{BIU,P}$ , can be represented by a Cassini oval with the BS and UE as its foci, and its equation is given by

$$(x^{2} + y^{2})^{2} - 2c^{2}(x^{2} - y^{2}) = d^{2}_{BIU,P} - c^{4},$$
 (3)

where the Cartesian coordinates of the typical UE and its serving BS are  $(\pm c, 0)$ , and (x, y) is the Cartesian coordinates of the IRSs. The enclosed area of the equal product-distance trajectory is determined by (4), as shown at the bottom of the next page, denoted by  $S(d_{\rm BIU,P})$ , where  $e_{\rm P} = \frac{\sqrt{d_{\rm BIU,P}}}{c}$  $K(\cdot)$  and  $E(\cdot)$  are the complete elliptic integral of the first and second kind, respectively [39].

Proof: See Appendix A.

*Remark 1:* The equal product-distance trajectory of the integrated path loss distance in Theorem 1 exhibits varying shapes depending on the parameter,  $e_{\rm P}$ , as shown in Fig. 3(a). In the following, we will discuss these four distinct shapes:

- If  $e_P < 1$ , i.e.,  $d_{BIU,P} < c^2$ , the trajectory consists of two foci-centered disconnected loops, the IRS with smallest  $d_{\rm BIUP}$  is supposed to be located in the vicinity of the BS or typical UE with a preference for being positioned between transceivers. This observation differs from the results obtained from the optimization perspective in [1]. In [1], the IRSs are constrained on the segment between the typical UE and its serving BS, within which the area outside the segment is not investigated, and it is a one-dimensional special case of this Cassini oval model.
- If  $e_P = 1$ , i.e.,  $d_{BIU,P} = c^2$ , the trajectory is a lemniscate of Bernoulli with the shape of a sideways figure eight and a double point at the middle-point of the BS and typical UE. The presence of double point in the trajectory causes the breakdown of area function in Theorem 1 into piecewise functions, producing an impulse on the PDF of  $d_{\rm BIU}$ , as verified numerically in Fig. 4(a).
- If  $e_P > 1$ , i.e.,  $d_{BIU,P} > c^2$ , the trajectory is a continuous loop. When  $1 < e_P < \sqrt{2}$ , the trajectory shapes like a peanut; When  $e_P > \sqrt{2}$ , the trajectory is a convex loop, and the vicinity preference of IRS vanishes, which is similar to the case of sum-distance path loss law.

Therefore, the individual link distance of the IRS→UE link  $d_{\rm IU}$  or BS $\rightarrow$ IRS link  $d_{\rm BI}$  does not solely determine the integrated path loss distance  $d_{\rm BIU}$ . As shown in Fig. 3, for both product- and sum-distance path loss laws,  $d_{\rm BIU}$  varies with the locations of the IRS for the same  $d_{IU}$  or  $d_{BI}$ . These geometric models offer robust theoretical support for the IRS deployment findings in [40] and [41], and elucidates the IRS's distinct behavior at varying elevations when the IRS is located between the BS and UE, which are also one-dimensional instances of the proposed Cassini oval model.

Using Theorem 1, we derived the PDF and CDF of the integrated path loss distance  $d_{\rm BIU}$  for the product-distance path loss model, as shown below.

Lemma 1: The CDF and PDF of the product distance  $d_{\text{BIU,P}}$  are respectively given by (5), as shown at the bottom of the next page, and (6), when IRS is located within the Cassini oval with maximum integrated path loss distance  $d_{t}$ ,

$$f_{\rm P}(d_{\rm BIU,P}) = \begin{cases} \frac{2d_{\rm BIU,P}}{c^2 S(d_{\rm t})} K\left(\frac{d_{\rm BIU,P}}{c^2}\right), & \text{for } e_{\rm P} < 1, \\ \frac{2}{S(d_{\rm t})} K\left(\frac{c^2}{d_{\rm BIU,P}}\right), & \text{for } e_{\rm P} \ge 1. \end{cases}$$

$$(6)$$

The enclosed area of  $d_t$ , denoted by  $S(d_t)$ , is defined in (4), which is based on the Cassini oval model in Theorem 1. 

*Proof:* See Appendix **B**.

## B. Sum-Distance Path Loss Law: Ellipse

 $\square$ 

The integrated path loss distance of the cascaded channel for the sum-distance path loss model is defined as the sum of individual link distance, i.e.,  $d_{BIU,S} = d_{BI} + d_{IU}$ , validated in [32] to be suitable for the near-field broadcasting scenario.

*Proposition 1 (Equal Sum-Distance Trajectory (Ellipse)):* For the sum-distance path loss law, the trajectory of IRSs with the equal integrated path loss distance of the cascaded BS $\rightarrow$ IRS $\rightarrow$ UE channel,  $d_{\text{BIU,S}}$ , follows an ellipse with the BS and UE as its foci, as expressed below

$$(\Delta - c^2) x^2 + \Delta y^2 = \Delta (\Delta - c^2), \quad \Delta \triangleq \frac{d_{\text{BIU,S}}^2}{4}, \quad (7)$$

where the Cartesian coordinates of the typical UE and its serving BS are  $(\pm c, 0)$ , and (x, y) is the Cartesian coordinates of the IRSs. The enclosed area of the equal sum-distance trajectory is determined by [42], denoted as  $S(d_{\text{BIU,S}})$ ,

$$S(d_{\rm BIU,S}) = \frac{\pi}{4} d_{\rm BIU,S} \sqrt{d_{\rm BIU,S}^2 - 4c^2}.$$
 (8)

Fig. 3(b) shows the equal sum-distance trajectory for sumdistance path loss model, where  $e_{\rm S} = \frac{2c}{d_{\rm BIU,S}}$ . The equal trajectories of the sum-distance path loss law are confocal ellipses. In the following Lemma, we derived the CDF and PDF of the integrated path loss distance for the sum-distance path loss law using the equations in Proposition 1.

Lemma 2: Based on Proposition 1, the CDF and PDF of the sum distance  $d_{\rm BIU,S}$  can be expressed as (9) and (10), respectively, when IRS is located within the Ellipse with maximum integrated path loss distance  $d_{t}$ .

$$F_{\rm S}(d_{\rm BIU,S}) = \frac{\pi d_{\rm BIU,S} \sqrt{d_{\rm BIU,S}^2 - 4c^2}}{4S(d_{\rm t})},\tag{9}$$



Fig. 3. Equal integrated path loss distance  $(d_{BIU})$  trajectories of IRS.

$$f_{\rm S}(d_{\rm BIU,S}) = \frac{\pi}{2S(d_{\rm t})} \frac{d_{\rm BIU,S}^2 - 2c^2}{\sqrt{d_{\rm BIU,S}^2 - 4c^2}}.$$
 (10)

The enclosed area of  $d_t$ , denoted by  $S(d_t)$ , is defined in (8).

#### C. Discussions on the Geometric Models

Building upon the geometric models, several observations have been made regarding the integrated path loss distance of the cascaded channel. First, the IRS locations with the smallest  $d_{\rm BIU}$  are dependent on both  $d_{\rm IU}$  and the relative positions of IRS and UE/BS, as shown in Fig. 3. Second, as demonstrated by the geometric models, there are instances where the preference for passive IRS location in the vicinity of transceivers is absent. Specifically, under product-distance path loss law with  $e_{\rm P} \ge 1$  and sum-distance path loss law, the preference of the vicinity of transceivers does not persist, which is different from the product-distance path loss law with  $e_{\rm P} < 1$ . Even though the product-distance path loss law maintains a preference for proximity if  $e_{\rm P} < 1$ , this preference is not uniform and diminishes with increasing  $e_{\rm P}$ .

The geometric models offer valuable insights into the deployment strategies of passive IRS. First, the conventional notion of deploying IRSs in the vicinity of transceivers is challenged, especially when the direct communication link is blocked. The reasons are as follows: a) The trajectories of



equal integrated path loss distances do not conform to uniform circular patterns with communication nodes as their centers. b) Environmental factors contribute to coherence issues, leading to compromised link performance when the IRS is located near transceivers due to the substantial blockage of the direct path. Second, the deployment strategies for IRS can be judiciously designed to achieve optimal utility while concurrently upholding a desirable outage probability. This is particularly relevant in scenarios involving fixed transmitter and receiver positions, such as M2M communication. Notably, the trajectories of equal integrated path loss distances are contingent upon their specific spatial placements, lending an inherent advantage to strategic deployment planning.

In multi-UE scenarios, such as [43] and [44], the IRS deployment strategy can be meticulously designed based on the geometric models. To illustrate the applications of the geometric models in deployment strategies, we introduce the concept of IRS effective zone for a transceiver pair characterized by the IRS location-based long-term performance.

Definition 3 (IRS Effective Zone for a Transceiver Pair): To ensure reliable communication between the typical UE and its serving BS, the maximum integrated path loss distance of the cascaded channel via passive IRS is defined as  $d_{\text{BIU,max}} = \left(\frac{P_{\text{T}}N^2\mathbb{E}[G_{\text{BIU}}]}{\tau\delta^2}\right)^{\frac{1}{\alpha}}$  to meet the outage signal-tonoise ratio (SNR) threshold,  $\tau$ , where  $\mathbb{E}[G_{\text{BIU}}]$  is the average

$$S(d_{\rm BIU,P}) = \begin{cases} 2c^2 \left[ E\left(\frac{d_{\rm BIU,P}}{c^2}\right) - \left(1 - \frac{d_{\rm BIU,P}^2}{c^4}\right) K\left(\frac{d_{\rm BIU,P}}{c^2}\right) \right], & \text{for } e_{\rm P} < 1\\ 2d_{\rm BIU,P} E\left(\frac{c^2}{d_{\rm BIU,P}}\right), & \text{for } e_{\rm P} \ge 1, \end{cases}$$
(4)

$$F_{\rm P}(d_{\rm BIU,P}) = \begin{cases} \frac{2c^2}{S(d_{\rm t})} \left[ E\left(\frac{d_{\rm BIU,P}}{c^2}\right) - \left(1 - \frac{d_{\rm BIU,P}^2}{c^4}\right) K\left(\frac{d_{\rm BIU,P}}{c^2}\right) \right], & \text{for } e_{\rm P} < 1\\ \frac{2d_{\rm BIU,P}}{S(d_{\rm t})} E\left(\frac{c^2}{d_{\rm BIU,P}}\right), & \text{for } e_{\rm P} \ge 1, \end{cases}$$
(5)

small-scale channel gain of the BS $\rightarrow$ IRS $\rightarrow$ UE channel. The area enclosed by the trajectory corresponding to  $d_{\rm BIU,max}$  is defined as the IRS effective zone for the transceiver pair.

As depicted in Fig. 1, the IRSs can be categorized into three types, based on the IRS effective zones of the transceiver pairs and the relative positions between the IRSs and transceivers. Therefore, in multi-UE scenarios, the association of IRS can be customized to various objective functions. For instance, considering the utility of IRS, we can opt for the Type-II IRS, capable of providing satisfactory performance for both UEs. Alternatively, concerning interference mitigation among UEs, the Type-I IRS may be selected, exclusively serving one UE.

## IV. OPTIMAL PLACEMENT AND OPPORTUNISTIC ASSOCIATION POLICY FOR HYBRID IRS

In this section, we first display the channel power statistics of the integrated IRS-aided cascaded channel, then determine the optimal placement of hybrid IRS, and last present an opportunistic association policy in terms of maximizing the long-term performance of the IRS-aided communication [45].

#### A. Channel Power Statistics

We assume that the phase shift of all reflecting elements are aligned to enhance the transmitted signals [8]. Thus, the received signal is given by

$$y = \left(\mathbf{g}_{\mathrm{BIU}}^{\mathrm{pas}} + \mathbf{g}_{\mathrm{BIU}}^{\mathrm{act}}\right) \epsilon d_{\mathrm{BIU}}^{-\alpha/2} x + \left(\mathbf{g}_{\mathrm{IU}}^{\mathrm{act}}\right)^{H} \Psi_{\mathrm{act}} \sqrt{\epsilon} d_{\mathrm{IU}}^{-\alpha/2} \mathbf{n}_{\mathrm{F}} + n_{0}, \qquad (11)$$

and the received SNR is given by

$$\mathbb{E}[\text{SNR}] = \mathbb{E}\left[\frac{P_{\text{T}}G_{\text{BIU}}d_{\text{BIU}}^{-\alpha}(N_{\text{pas}} + \eta N_{\text{act}})^2}{N_{\text{act}}\eta^2 \delta_{\text{F}}^2 G_{\text{IU}}d_{\text{IU}}^{-\alpha} + \delta^2}\right],\qquad(12)$$

where  $\mathbf{n}_{\rm F} \in \mathbb{C}^{N_{\rm act} \times 1}$  is the thermal noise generated by the active part, which is assumed to follow the Complex Normal distribution, i.e.,  $\mathbf{n}_{\rm F} \sim \mathcal{CN}(\mathbf{0}_{N_{\rm act}}, \delta_{\rm F}^2 \mathbf{I}_{N_{\rm act}})$ ,  $n_0$  is the additive white Gaussian noise (AWGN) at the user, i.e.,  $n_0 \sim \mathcal{N}(0, \delta_0^2)$ , and  $G_{\rm BIU}$  ( $G_{\rm IU}$ ) is the small-scale channel gain of the BS $\rightarrow$ IRS $\rightarrow$ UE (IRS $\rightarrow$ UE) channel.

Herein, for expression concise, we assume all reflecting elements suffer the same small-scale fading, and all transmission links follow Nakagami-m fading, i.e.,  $|g_{\rm BI}|$  and  $|g_{\rm IU}|$ follow Nakagami-m distributions with parameters  $m_{\rm BI}$  and  $m_{\rm IU}$ , respectively. The analysis can be extended to generalized fading straightforwardly as the small-scale fading is assumed independent of the link distance.

Given Nakagami-*m* distributed fading, the channel gain of BS $\rightarrow$ IRS and IRS $\rightarrow$ UE links can be represented as Gamma distributions with unit mean. The distribution of small-scale channel gain for individual links, i.e., BS $\rightarrow$ IRS and IRS $\rightarrow$ UE, are given by  $G_{\rm BI} \sim \mathcal{MG}(\varepsilon_{\rm BI}, \beta_{\rm BI}, \xi_{\rm BI})$  and  $G_{\rm IU} \sim \mathcal{MG}(\varepsilon_{\rm IU}, \beta_{\rm IU}, \xi_{\rm IU})$  with I = 1, where

$$\varepsilon_{\rm BI} = \frac{m_{\rm BI}^{m_{\rm BI}}}{\Gamma(m_{\rm BI})}, \quad \beta_{\rm BI} = \xi_{\rm BI} = m_{\rm BI},$$
  

$$\varepsilon_{\rm IU} = \frac{m_{\rm IU}^{m_{\rm IU}}}{\Gamma(m_{\rm IU})}, \quad \beta_{\rm IU} = \xi_{\rm IU} = m_{\rm IU}.$$
(13)

Moreover, the multiplicability of the mixture Gamma distribution, as proven in [9], allows for modeling the integrated channel gain of the cascaded BS $\rightarrow$ IRS $\rightarrow$ UE channel as a mixture Gamma distribution, denoted as  $G_{\text{BIU}} \triangleq G_{\text{BI}}G_{\text{IU}} \sim \mathcal{MG}(\varepsilon_{\text{BIU},i}, \beta_{\text{BIU},i}, \xi_{\text{BIU},i})$ , and the parameters are given by

$$\varepsilon_{\mathrm{BIU},i} = \frac{(m_{\mathrm{BI}}m_{\mathrm{IU}})^{m_{\mathrm{BI}}}w_{i}t_{i}^{m_{\mathrm{IU}}-m_{\mathrm{BI}}-1}}{\Gamma(m_{\mathrm{BI}})\Gamma(m_{\mathrm{IU}})},$$
  
$$\beta_{\mathrm{BIU},i} = m_{\mathrm{BI}}, \quad \xi_{\mathrm{BIU},i} = \frac{m_{\mathrm{BI}}m_{\mathrm{IU}}}{t_{i}}, \qquad (14)$$

where  $t_i$  is the *i*-th root of the Laguerre polynomial  $L_p(t)$ , and  $w_i$  is the *i*-th weight of the Gaussian-Laguerre quadrature  $\int_0^\infty e^{-t} f(t) dt \approx \sum_{i=1}^I w_i f(t_i)$  that is defined as  $w_i = \frac{t_i}{(p+1)^2 L_{p+1}(t_i)^2}$  [39].

## B. Optimal Locations of the Hybrid IRS

To determine the optimal locations of the hybrid IRS, taking into account the long-term average received SNR, the problem is formulated as below

(P1): 
$$\max_{d_{\mathrm{BI}}, d_{\mathrm{IU}}} \mathbb{E}[\mathrm{SNR}]$$
  
s.t.  $P_{\mathrm{T}} \ge 0, P_{\mathrm{F}} \ge 0,$  (15a)

$$\eta^2 \le \frac{P_{\rm F}}{N_{\rm act}(P_{\rm T}G_{\rm BI}d_{\rm BI}^{-\alpha} + \delta_{\rm F}^2)},\qquad(15b)$$

$$P_{\rm F} \ge N_{\rm act} (P_{\rm T} G_{\rm BI} d_{\rm BI}^{-\alpha} + \delta_{\rm F}^2), \qquad (15c)$$

$$d_{\rm BIU} = d_{\rm BI} \circ d_{\rm IU},\tag{15d}$$

$$d_{\rm BI} \ge 0, \quad d_{\rm IU} \ge 0, \tag{15e}$$

where (15b) and (15c) represent the hardware limit of the active reflecting elements and the power constraint for the amplification factor [36]. Note that, the inequality in (15b) holds for general cases, including dynamic  $d_{\rm BI}$ , and the equal symbol holds for associated IRS. Furthermore, (15c) indicates  $\eta \ge 1$  [19].

The optimization in (P1) is a challenging problem due to the coupled relation between the thermal noise and the amplitude amplification factor. The thermal noise is generated by active parts and the amplification factor depends on the channel condition of the BS $\rightarrow$ IRS link, which affects both the received signal power and thermal noise at the UE side. To simplify this problem, we note that (P1) includes the conventional IRS architectures as two special cases. Specifically, if  $N_{\rm act} = 0$ , the hybrid IRS reduces to the conventional passive IRS, and the solution of (P1) is the optimal locations of passive IRS. Otherwise, the object of (P1) is to find the optimal locations of the active IRS when  $N_{\rm pas} = 0$ .

Thereby, we opt to initially partition this problem into two distinct parts, aiming to ascertain the optimal locations for the passive and active sub-surfaces individually. The optimal locations for the passive sub-surface are determined as points along the trajectory possessing the smallest value of  $d_{\rm BIU}$ , where each  $d_{\rm BIU}$  represents a trajectory consisting of potential IRS locations that yield identical performance without considering small-scale fading and blockage. In contrast, performance of the active sub-surface varies along this trajectory due to the amplification factor and the additional thermal noise generated by the active parts.

Hence, our initial focus lies on determining the optimal locations of the active sub-surface on a given trajectory defined by  $d_{\text{BIU}}$ , and the problem is formulated as follows

(P2): 
$$\max_{d_{\rm BI}, d_{\rm IU}} \mathbb{E}[{\rm SNR}]^{(\rm act)}$$
 s.t.  $(15a) - (15e),$ 

where  $\mathbb{E}[SNR]^{(act)}$  is given by

$$\mathbb{E}[\text{SNR}]^{(\text{act})} = \mathbb{E}\left[\frac{P_{\text{T}}G_{\text{BIU}}d_{\text{BIU}}^{-\alpha}\eta^2 N_{\text{act}}^2}{N_{\text{act}}\eta^2 \delta_{\text{F}}^2 G_{\text{IU}}d_{\text{IU}}^{-\alpha} + \delta^2}\right].$$
 (16)

Then, with the maximum value of the amplification factor,  $\mathbb{E}[\text{SNR}]^{(\text{act})}$  can be derived as (17), as shown at the bottom of the next page, where the mean value of the moments of the small-scale fading can be derived by substituting (13), (14) and (2c), as described below<sup>4</sup>

$$\mathbb{E}\left[G_{\rm BI}^{-1}\right] = \frac{m_{\rm BI}}{m_{\rm BI} - 1}, \quad \mathbb{E}\left[G_{\rm IU}^{-1}\right] = \frac{m_{\rm IU}}{m_{\rm IU} - 1},$$
$$\mathbb{E}\left[G_{\rm BIU}^{-1}\right] = \sum_{i=1}^{I} \frac{m_{\rm BI} m_{\rm IU} w_i t_i^{m_{\rm IU} - 1}}{(m_{\rm BI} - 1)\Gamma(m_{\rm IU})}.$$
(18)

Thus, (P2) can be further simplified as

(P3): 
$$\min_{d_{\rm BI}, d_{\rm IU}} f = P_{\rm F} \delta_{\rm F}^2 d_{\rm BI}^{\alpha} \mathbb{E} \left[ G_{\rm BI}^{-1} \right] + P_{\rm T} \delta^2 d_{\rm IU}^{\alpha} \mathbb{E} \left[ G_{\rm IU}^{-1} \right]$$
  
s.t. (15*a*), (15*c*) - (15*e*).

Next, by employing the two path loss laws, the results are achieved as below,<sup>5</sup>

$$d_{\rm BI}^{*} = \rho \sqrt{d_{\rm BIU}}, \quad \text{for product},$$
  
$$d_{\rm BI}^{*} = \frac{d_{\rm BIU}}{\varphi_{\rm S} + 1}, \quad \text{for sum},$$
 (19)

where  $\rho = \left(\frac{P_{\rm T} \delta^2 \mathbb{E}[G_{\rm IU}^{-1}]}{P_{\rm F} \delta_{\rm F}^2 \mathbb{E}[G_{\rm BI}^{-1}]}\right)^{\frac{1}{2\alpha}}$ , and  $\varphi_{\rm S} = \rho^{\frac{2\alpha}{1-\alpha}}$ . Thereby, the corresponding distance between IRS and UE can be readily given by

$$\begin{aligned} d_{\rm IU}^* &= \rho^{-1} \sqrt{d_{\rm BIU}}, & \text{for product,} \\ d_{\rm IU}^* &= d_{\rm BIU} - (d_{\rm BI})^*, & \text{for sum.} \end{aligned}$$
(20)

It can be verified that (P3) is feasible if and only if the following constraints of system parameters are satisfied [21].

$$\begin{split} \rho &\geq \left(\frac{N_{\rm act}P_{\rm T}}{P_{\rm F} - N_{\rm act}\delta_{\rm F}^2}\right)^{\frac{1}{\alpha}} d_{\rm BIU,P}^{-\frac{1}{2}}, \qquad \text{for product,} \\ \varphi_{\rm S} &\leq \left(\frac{N_{\rm act}P_{\rm T}}{P_{\rm F} - N_{\rm act}\delta_{\rm F}^2}\right)^{-\frac{1}{\alpha}} d_{\rm BIU,S} - 1, \quad \text{for sum.} \quad (21) \end{split}$$

As illustrated in Fig. 3(a), the feasibility of the local optimal points for product-distance path loss law in (19) varies along  $e_{\rm P}$ , due to the various trajectory shapes. Specifically, the

<sup>4</sup>Note that in this study, we employ Nakagami-m fading for expression conciseness. It is worth mentioning that the outcomes derived from our approach can be extended to generalized fading, which is versatile for application across various frequency bands [37].

<sup>5</sup>For the product-distance path loss law, when the power and noise level of both the BS and the active IRS are comparable, the optimal solution lies at the line that is perpendicular to and divides the segment made by BS and UE into two equal halves. This observation is consistent with relay systems. Besides, the feasibility conditions can offer theoretical backing for system design as well.

optimal points for product-distance path loss law on each trajectory are available if and only if the system parameters are designed based on the following inequality

$$\rho + \frac{1}{\rho} \ge \frac{2}{e_{\rm P}}.\tag{22}$$

If  $e_P \ge 1$ , the inequality in (22) is always satisfied and the results in (19) are robust. However, if  $e_P < 1$ , the value of  $d_{BI}^*$  may be infeasible in certain parameter configurations, i.e.,  $\rho + 1/\rho < 2/e_P$ . Then, the minimum  $d_{BI}$  on each trajectory is sub-optimal, considering that the function f is convex and active IRS is supposed to be deployed near UE instead of BS due to practical operation power of BS and IRS [21]. Intuitively, as  $e_P \rightarrow 0$ , the active IRS is supposed to be deployed to be deployed in the vicinity of receivers and it should operate at lower power mode.

*Remark 2:* The optimal locations of the active sub-surface along a predetermined trajectory with a fixed  $d_{\rm BIU}$  rely on  $\rho$  and  $\alpha$ . This finding is consistent with [21], where the active IRS is confined to the segment between UE and BS with limited height.

In the following, we assume that the system is designed based on (21) and (22). With the results in (19), (20) and the geometric models studied in Section III, the optimal locations of the active sub-surface are derived in the following lemma.

Lemma 3: When the typical UE and its serving BS are located at  $(\pm d_{\rm BU}/2, 0)$ , the coordinates of the optimal locations of the active sub-surface are given by

$$x_{1}^{*} = \frac{\varphi_{\rm P}^{2} - d_{\rm BIU}^{2}}{2d_{\rm BU}\varphi_{\rm P}}, \qquad y_{1}^{*} = \pm \sqrt{\varphi_{\rm P} - (x^{*} + c)^{2}},$$
$$x_{2}^{*} = \frac{d_{\rm BU}^{2} - d_{\rm BIU}^{2} \left(1 - \frac{2}{\varphi_{\rm S}}\right)}{2d_{\rm BU}}, \quad y_{2}^{*} = \pm \sqrt{\left(\frac{d_{\rm BIU}}{\varphi_{\rm S} + 1}\right)^{2} - (x^{*})^{2}},$$
(23)

where  $\varphi_{\rm P} = \rho^2 d_{\rm BIU}$ ,  $(x_1^*, y_1^*)$  and  $(x_2^*, y_2^*)$  are the optimal coordinates of the active sub-surface for the product- and sum-distance path loss laws, respectively.

*Proof:* See Appendix C.

However, the results in Lemma 3 are local solutions for a predetermined  $d_{\rm BIU}$ . Subsequently, we proceed to acquire the globally optimal positions of active sub-surface by searching all the local optimal solutions across varying  $d_{\rm BIU}$ , the problem of determining the globally optimal solutions for (P2) is formulated as<sup>6</sup>

(P4): 
$$\max_{d_{\rm BIU}} \frac{P_{\rm T} P_{\rm F} N_{\rm act}^2 d_{\rm BIU}^{-\alpha}}{P_{\rm F} \delta_{\rm F}^2 (d_{\rm IU}^*)^{-\alpha} + P_{\rm T} \delta^2 (d_{\rm BI}^*)^{-\alpha} + \delta^2 \delta_{\rm F}^2}$$
  
s.t. (15*a*), (15*d*) - (15*e*), (19), (20).

Next, we applied the results in (19) and (20) to (P4), which means the optimal locations of IRS for each determined  $d_{\rm BIU}$  are selected. When the product-distance path loss law is adopted, (P4) can be simplified to

(P5): 
$$\max_{d_{\rm BIU}} \frac{P_{\rm T} P_{\rm F} N_{\rm act}}{2\sqrt{P_{\rm F} P_{\rm T} \delta_{\rm F}^2 \delta^2} d_{\rm BIU}^{\frac{\alpha}{2}} + \delta^2 \delta_{\rm F}^2 d_{\rm BIU}^{\alpha}}$$

<sup>6</sup>Because the small-scale fading has been incorporated in the optimal solutions in (19) and (20) as a scaling factor of the path loss, the following objective functions omit the small-scale fading parameter [46].

When the sum-distance path loss law is adopted, (P4) can be simplified to

(P6) :max  

$$_{d_{\mathrm{BIU}}} \frac{P_{\mathrm{T}}P_{\mathrm{F}}N_{\mathrm{act}}}{P_{\mathrm{F}}\delta_{\mathrm{F}}^{2}\left(\frac{\varphi_{\mathrm{S}}}{\varphi_{\mathrm{S}}+1}\right)^{-\alpha} + P_{\mathrm{T}}\delta^{2}\left(\frac{1}{\varphi_{\mathrm{S}}+1}\right)^{-\alpha} + \delta_{\mathrm{F}}^{2}d_{\mathrm{BIU}}^{\alpha}}$$

The singularity of both (P5) and (P6) with respect to  $d_{\rm BIU}$  clearly indicates that the optimal solutions exist at the minimum value of  $d_{\rm BIU}$  while still satisfying the condition for optimal placement of the passive sub-surface. Therefore, the optimal locations of the hybrid IRS are given by

$$d_{\rm BI}^{**} = \rho \sqrt{d_{\rm BIU,min}}, \ d_{\rm IU}^{**} = \rho^{-1} \sqrt{d_{\rm BIU,min}}, \ \text{ for product}, d_{\rm BI}^{**} = \frac{d_{\rm BIU,min}}{\varphi_{\rm S} + 1}, \qquad d_{\rm IU}^{**} = d_{\rm BIU,min} - d_{\rm BI}^{**}, \ \text{for sum.} \ (24)$$

*Remark 3:* The optimal locations for passive, active, and hybrid IRS under both sum- and product-distance path loss laws are determined by the integrated path loss distance of the cascaded BS $\rightarrow$ IRS $\rightarrow$ UE channel, small-scale fading of each link, and network parameters, such as  $P_{\rm T}$ ,  $P_{\rm F}$ ,  $\delta^2$ ,  $\delta^2_{\rm F}$ , and  $\alpha$ .

### C. An Opportunistic Association Policy for the Hybrid IRS

Based on the obtained optimal placement, we propose a two-step integrated path loss distance-based opportunistic association policy for the hybrid IRS. Assuming that the hybrid IRSs are densely deployed throughout the entire area, and the optimally located hybrid IRSs on the trajectories for all  $d_{\rm BIU}$  are available, whose Cartesian coordinates are given by Lemma 3. The association procedures are defined as follows:

- 1) Determine the hybrid IRSs with the minimum integrated path loss distance of the cascaded channel,  $d_{\rm BIU,min}$ , according to the practical scenario.
- 2) Select the hybrid IRS located at the optimal locations along the trajectory defined by  $d_{\rm BIU,min}$  based on the results in Lemma 3.

In practical scenarios, the locations of BSs and IRSs are typically static. Consequently, when a UE connects to the network, it can report its location information to the BSs, and then the BSs have the capability to generate a lookup table for the UE, presenting both the index and the integrated path loss distance pertaining to potential IRSs. As such, both the BS and UE can determine the IRS with minimum integrated path loss distance.

#### V. PERFORMANCE ANALYSIS

When the serving IRS is positioned optimally, the average received SNR of hybrid IRS-aided communication is expressed as

$$\mathbb{E}[\text{SNR}] = \frac{P_{\text{T}} d_{\text{BIU,min}}^{-\alpha} (N_{\text{pas}} + \eta^{**} N_{\text{act}})^2}{N_{\text{act}} (\eta^{**})^2 \delta_{\text{F}}^2 (d_{\text{IU}}^{**})^{-\alpha} + \delta^2}, \qquad (25)$$

where the optimal amplification factor is obtained by substituting (24) into (P1), as shown below

$$(\eta^{**})^{2} = \begin{cases} \frac{P_{\rm F}}{N_{\rm act} \left(P_{\rm T} \rho^{-\alpha} d_{\rm BIU,\min}^{-\frac{\alpha}{2}} + \delta_{\rm F}^{2}\right)}, & \text{for product,} \\ \frac{P_{\rm F}}{N_{\rm act} \left(P_{\rm T} \left(\frac{d_{\rm BIU,\min}}{\varphi_{\rm S}+1}\right)^{-\alpha} + \delta_{\rm F}^{2}\right)}, & \text{for sum .} \end{cases}$$

$$(26)$$

From (25) and (26), it is observed that the distinction between hybrid and passive IRS-aided communication analysis lies in the amplification factor and the thermal noise introduced by the active parts. In addition, with the assistance of the geometric models, the average received SNR can be expressed as a single variable function with respect to  $d_{\rm BIU}$ .

However, the coupling between the numerator and denominator of SNR resulting from the active sub-surface still leads to intractable analysis. Due to page limits, we leave this into our future work, and focus on exploring benefits of the geometric models and opportunistic association policy introduced in this work. To this end, we study a special case in the subsequent analysis, where  $N_{\rm act} = 0$  and the hybrid IRS is transformed into a passive IRS.<sup>7</sup>

It is noteworthy that the existing research on the analysis of passive IRS-aided networks is still constrained by the challenge of modeling the integrated IRS-aided cascaded channel [7], [9]. While the integrated small-scale fading has been accurately and tractably modeled as a mixture Gamma distribution in [9], the precise modeling of the integrated path loss distance remains unsolved. The geometric models examined in this work effectively address this issue and facilitate the modeling of the entire integrated IRS-aided cascaded channel, employing an opportunistic association policy.

We study two cases in this section, namely communications assisted by a single IRS or multiple IRSs. We consider an edge-user as the typical UE and assume that the direct link between the typical UE and its serving BS is blocked due to a long transmission distance.

## A. Single IRS-Aided Network

In this subsection, we will delve into single IRS-aided communication, where the signal received by the typical UE from the connected IRS is

$$y = \epsilon d_{\rm BIU}^{-\alpha/2} \mathbf{g}_{\rm BIU} x + n_0, \tag{27}$$

where x is the transmitted signal with power  $P_{\rm T}$ . Then, the SNR is given by

$$SNR = \frac{P_{\rm T} N^2 \mathbb{E} \left[ G_{\rm BIU} \right] d_{\rm BIU}^{-\alpha}}{\delta^2}.$$
 (28)

<sup>7</sup>We will investigate the network-level performance of active sub-surface aided systems in our upcoming research.

$$\mathbb{E}[\mathrm{SNR}]^{(\mathrm{act})} = \frac{P_{\mathrm{T}} P_{\mathrm{F}} N_{\mathrm{act}} d_{\mathrm{BIU}}^{-\alpha}}{P_{\mathrm{F}} \delta_{\mathrm{F}}^2 d_{\mathrm{IU}}^{-\alpha} \mathbb{E}\left[G_{\mathrm{BI}}^{-1}\right] + P_{\mathrm{T}} \delta^2 d_{\mathrm{BI}}^{-\alpha} \mathbb{E}\left[G_{\mathrm{IU}}^{-1}\right] + \delta^2 \delta_{\mathrm{F}}^2 \mathbb{E}\left[G_{\mathrm{BIU}}^{-1}\right]},\tag{17}$$

1) Outage Probability: The outage probability is given by

$$\mathcal{P}_{\text{out}} = \mathcal{P}\left(\text{SNR} < \tau\right) = \mathcal{P}\left(d_{\text{BIU}} > \left(\frac{P_{\text{T}}N^2\mathbb{E}\left[G_{\text{BIU}}\right]}{\delta^2\tau}\right)^{\frac{1}{\alpha}}\right).$$
(29)

Then, by substituting the CDF expression of the integrated path loss distance under different path loss laws, (29) can be resolved. When the sum-distance path loss law is adopted, the outage probability is given by

$$\mathcal{P}_{\text{out}} = 1 - \frac{\pi}{4S(d_{\text{t}})} \varrho \tau^{-\frac{1}{\alpha}} \sqrt{\varrho^2 \tau^{-\frac{2}{\alpha}} - 4c^2}, \qquad (30)$$

where  $\rho = \left(\frac{P_{\mathrm{T}}N^2\mathbb{E}[G_{\mathrm{BIU}}]}{\delta^2}\right)^{\frac{1}{\alpha}}$ . When the product-distance path loss law is adopted, the outage probability is given by

$$\mathcal{P}_{\text{out}} = \begin{cases} 1 - \frac{2c^2}{S(d_{\text{t}})} \left[ E\left(\varpi\right) - \left(1 - \varpi^2\right) K\left(\varpi\right) \right], & \text{for } e_{\text{P}} < 1\\ 1 - \frac{2\varpi c^2}{S(d_{\text{t}})} E\left(\varpi^{-1}\right), & \text{for } e_{\text{P}} \ge 1, \end{cases}$$
(31)

where  $\varpi = \frac{\varrho}{c^2 \tau^{\frac{1}{\alpha}}}$ . 2) *Low-Rate Probability:* The low-rate probability is defined as the probability that the rate is lower than the threshold,  $r_0$ , which is given by

$$\mathcal{P}_{\mathrm{R}} = \mathcal{P}\left(\ln(1 + \mathrm{SNR}) < r_{0}\right)$$
$$= \mathcal{P}\left(d_{\mathrm{BIU}} > \left[\frac{P_{\mathrm{T}}N^{2}\mathbb{E}\left[G_{\mathrm{BIU}}\right]}{\delta^{2}(e^{r_{0}} - 1)}\right]^{\frac{1}{\alpha}}\right), \qquad (32)$$

which can be expressed by the CDF of the integrated path loss distance as well.

3) Relationship Between Outage Probability and Low-Rate Probability: By comparing (29) and (32), the low-rate probability can be expressed as a function of the outage probability

$$\mathcal{P}_{\mathrm{R}} = \overline{F}\left[\left(\frac{\tau}{e^{r_0} - 1}\right)^{\frac{1}{\alpha}} \overline{F}^{-1}(\mathcal{P}_{\mathrm{out}})\right],\tag{33}$$

where  $\overline{F}(\cdot) = 1 - F(\cdot)$ , and  $F(\cdot)$  is the CDF of the integrated path loss distance, given in (5) and (9).

## B. Multiple IRSs-Aided Communication

In this subsection, we study the multiple IRSs-aided communication, where one reference cascaded link is selected, and the other IRSs align their phase shift with the reference link to corporately enhance the communication between transceivers.

1) Association Policy: To ensure effective operation within networks assisted by multiple IRSs, it is crucial to establish a suitable association policy for the IRSs.

As shown in Definition 3, the typical UE associates to its serving BS through IRSs positioned within the IRS effective zone for  $UE_0$  and  $BS_0$ , as stated below.

UE<sub>0</sub> associates with BS<sub>0</sub> through a group of IRSs {IRS<sub>l</sub>}  

$$\Leftrightarrow$$
 {IRS<sub>l</sub>} =  $\arg_{\text{IRS}_l \in \mathcal{J}} d_{\text{BIU},l} \leq d_{\text{BIU},\text{max}}$ ,

where  $l = 1, \dots, L, L$  is the total number of the connected IRSs. As the IRSs are uniformly distributed throughout the whole area, the number of IRSs located within the IRS effective zone is given by

$$L = \lambda_{\rm I} \cdot S(d_{\rm BIU,max}) = \lambda_{\rm I} \cdot S\left(\varrho \tau^{-\frac{1}{\alpha}}\right), \qquad (34)$$

where  $\lambda_{\rm I} = J/S(d_{\rm t})$  is the density of IRSs,  $S(d_{\rm BIU,max})$ is the area of the IRS effective zone under the threshold,  $\tau$ , which depends on the selected path loss model, as shown in (4) and (8).

Thus, the received signal at the typical UE is given by

$$y = \sum_{l=1}^{L} h_{\text{BIU},l} x + n_0,$$
(35)

where  $h_{\text{BIU},l}$ ,  $l = 1, \dots, L$ , is the channel of the *l*-th cascaded BS $\rightarrow$ IRS<sub>*l*</sub> $\rightarrow$ UE link.

When the serving IRSs are located at the same equal gain trajectory, the average received signal power is given by

$$\mathbb{E}[P_{\rm r}] = L^2 N^2 P_{\rm T} \epsilon^2 d_{\rm BIU}^{-\alpha} \mathbb{E}[G_{\rm BIU}].$$
(36)

Then, the average received SNR is given by

$$\mathbb{E}[\text{SNR}] = L^2 \mathbb{E}[\text{SNR}_l], \quad \mathbb{E}[\text{SNR}_l] = N^2 \frac{P_{\text{T}} \epsilon^2 d_{\text{BIU}}^{-\alpha} \mathbb{E}[G_{\text{BIU}}]}{\delta^2}.$$

It is observed that the phase shift alignment among multiple associated IRSs provides a SNR gain in order of  $O(L^2N^2)$ , which is consistent with a centralized IRS with LN reflecting elements.

2) Channel Statistics: To evaluate the performance metrics, we focus on the mixture channel coefficient  $|h_{\rm S}| \triangleq$  $\left|\sum_{l=1}^{L} h_{\mathrm{BIU},l}\right|.$ 

Proposition 2: Given each single link follows generalized fading distribution, the amplitude of each cascaded channel,  $|h_{\text{BIU},l}|$ , can be modeled as a mixture Nakagami-m distribution, whose PDF is given by

$$f_{\rm X}(x) = \sum_{i=1}^{I} 2\varepsilon_i x^{2\beta_i - 1} e^{-\xi_i x^2},$$
(37)

where I is the number of Nakagami-m terms, which is set as 17 to achieve an error less than  $10^{-5}$ ,  $\{\varepsilon_i, \beta_i, \xi_i\}$  is the parameters of each term [39].

*Proof:* See Appendix D. 
$$\Box$$

Lemma 4: The amplitude of the mixture channel coefficient can be modeled as a Gaussian distributed random variable, denoted as  $|h_{\rm S}| \sim \mathcal{N}(\mu, \delta_{\rm S})$ , where  $\mu$  and  $\delta_{\rm S}$  are given by

$$\mu = \sum_{l=1}^{L} \mu_l, \quad \delta_{\rm S}^2 = L - \sum_{l=0}^{L} \mu_l^2, \tag{38}$$

$$\mu_{l} = \sum_{i=1}^{I} \varepsilon_{i}^{(l)} \left( \beta_{i}^{(l)} - \frac{1}{2} \right) ! \xi_{i}^{(l) - \beta_{i}^{(l)} - \frac{1}{2}}, l = 1, \cdots, L, \quad (39)$$

and  $\{\varepsilon_i^{(l)}, \beta_i^{(l)}, \xi_i^{(l)}\}$  is the distribution parameters of the channel coefficient of the l-th serving IRS-aided cascaded link  $(BS_0 \rightarrow IRS_l \rightarrow UE_0)$  [39].

*Proof:* See Appendix E.  Proposition 3: When the serving IRSs are located on the equal trajectory with identical  $d_{\rm BIU}$ , and each IRS-aided channel suffers the same small-scale fading, the received signal power can be modeled as  $|h_{\rm S}| \sim \mathcal{N}(\mu, \delta_{\rm S}^2)$ , where  $\mu$  and  $\delta_{\rm S}^2$  are

$$\mu = L\mu_l, \quad \delta_{\rm S}^2 = L - L\mu_l^2.$$
 (40)

3) *Received Signal Power:* As the serving IRSs align phase shifts with a reference link, the received signal power is given by

$$S = P_{\rm T} \left| \sum_{l=1}^{L} h_{\rm BIU,l} \right|^2.$$
(41)

Then, we model the received signal power as a Gamma distribution with the assistance of the second and fourth moments of amplitude of the channel coefficients, denoted as  $S \sim \text{Gamma}(k, \theta)$  with parameters [7]

$$k = \frac{\left(\mathbb{E}[|h_{\rm S}|^2]\right)^2}{\mathbb{E}[|h_{\rm S}|^4] - \left(\mathbb{E}[|h_{\rm S}|^2]\right)^2}, \quad \theta = \frac{\mathbb{E}[|h_{\rm S}|^4] - \left(\mathbb{E}[|h_{\rm S}|^2]\right)^2}{\mathbb{E}[|h_{\rm S}|^2]},$$
(42)

where  $\mathbb{E}[|h_{\rm S}|^2] = \delta_{\rm S}^2 + \mu^2$ , and  $\mathbb{E}[|h_{\rm S}|^4] = 3\delta_{\rm S}^4 + 6\delta_{\rm S}^2\mu^2 + \mu^4$ .

As the received signal power is modeled as a Gamma distribution, denoted as  $S \sim \text{Gamma}(k, \theta)$ , the system performance can be evaluated using the tools in [47], where all the performance metrics are expressed as functions of SNR [9], [46].

4) Spectral Efficiency: Spectral efficiency is defined as

$$\mathcal{R} = \mathbb{E}[\ln(1 + \text{SNR})]. \tag{43}$$

By utilizing the distribution parameters of the signal power in (42), the spectral efficiency can be derived by utilizing the analysis framework proposed in [47].

$$\mathcal{R} = \int_0^\infty \frac{1}{z} \left( 1 - \frac{1}{(1+z)^k} \right) e^{-\frac{\delta^2 z}{\theta}} \mathrm{d}z.$$
(44)

5) *Moments of SNR:* The moments of the SNR  $\mathbb{E}[SNR^r]$  can be evaluated as follows [46]

$$\mathbb{E}[\mathrm{SNR}^r] = \int_0^\infty \frac{\Gamma(k+r)}{\Gamma(k)\Gamma(r)} z^{r-1} e^{-\frac{\delta^2}{\theta}z} \mathrm{d}z.$$
(45)

## VI. NUMERICAL RESULTS

In this section, Monte-Carlo (MC) simulation with  $10^6$  iterations is carried out using MATLAB to verify the analysis and compare the performance under different system setups. The simulation setup is as follows, if not specified otherwise. The path loss exponent is  $\alpha = 3$ , the transmit power of the BS is  $P_{\rm T} = 20$  dBm, the amplification power of the active reflecting elements is  $P_{\rm F} = 5$  dBm, the noise power at the receiver is  $\delta^2 = -90$  dBm, the thermal noise power generated by the active reflecting element is  $\delta_{\rm F}^2 = -80$  dBm, the half of  $d_{\rm BU}$  is c = 30 m, the number of passive IRS elements is  $N_{\rm pas} = 300$ , the number of active reflecting elements is  $N_{\rm act} = 40$ , and the carrier frequency is 2 GHz [19], [36].

#### A. Distribution of the Integrated Path Loss Distance

We first display the PDF of the integrated path loss distance,  $d_{\rm BIU}$ , with different  $d_{\rm BU}$  for both product- and sum-distance path loss laws in Fig. 4, thereby validating the findings in Lemma 1 and Lemma 2.

It has been observed that the PDF of the integrated path loss distance scales with the link distance between BS and UE. The results align with the geometric models depicted in Fig. 3, indicating that the distribution of  $d_{\rm BIU}$  is dependent on  $d_{\rm BU}$ . Consequently, even in instances where the direct link is blocked, a BS in closer proximity of the typical UE, as indicated by a smaller  $d_{\rm BU}$ , may have a higher probability of delivering superior performance.

In Fig. 4(a), for product-distance path loss law, it is observed that the PDF exhibits an initial increase followed by a subsequent decrease and the demarcation point is  $d_{\rm BIU,P} = d_{\rm BU}^2/4$ , i.e.,  $e_{\rm P} = 1$ . On the other hand, an opposite trend is observed in the sum-distance path loss law and the demarcation point is  $d_{\rm BIU,S} = \sqrt{3/2} \ d_{\rm BU}$ , i.e.,  $e_{\rm S} = \sqrt{2/3}$ , as displayed in Fig. 4(b). By comparing Figs. 4(a) and 4(b), it can be observed that the system suffers severer degradation when employing the product-distance path loss law.

## B. Achievable Rate Under Different Path Loss Laws

The three-dimensional (3D) simulation of the achievable rate in the IRS-aided communication is presented in Fig. 5, showcasing the impact of different path loss laws for passive/hybrid IRS.

As shown in Figs. 5(a) and 5(b), when the IRS is fully passive, the performance has strong geometric symmetry, which aligns with the geometric models of the integrated path loss for the cascaded channels introduced in Section III. This is expected since the received SNR of passive IRS-aided communication is proportional to the channel gain of the integrated cascaded channel.

However, the adoption of the hybrid IRS breaks the symmetry between the UE side and BS side, owing to the location preference of the active IRS sub-surface, as shown in Figs. 5(c) and 5(d). Furthermore, the optimal locations of the hybrid IRS depend on the integrated path loss distance of the cascaded channel and the system parameters, such as the transmit power of the serving BS  $P_{\rm T}$ , the amplification power of the active reflecting elements  $P_{\rm F}$ , the power of the thermal noise introduced by the active parts  $\delta_{\rm F}^2$ , the thermal noise at the typical UE  $\delta^2$ , and the path loss exponent  $\alpha$ .

Moreover, compared to the product-distance path loss law, the impact of IRS locations is relatively insignificant when adopting the sum-distance path loss law, as shown in Fig. 5(d). For example, in the case of passive IRS, the achievable rate gap that can be achieved across the entire area is about 15 nats/sec/Hz for the product-distance path loss law, while it is around 3 nats/sec/Hz for the sum-distance path loss law.

#### C. Nearest and Opportunistic IRS Association Policy

In Fig. 6, we present the average received SNR from IRS-aided channels versus the IRS density for both product-/sum-distance path loss laws. The 'o' and 'n' represent



(c) Product-distance path loss law (Hybrid IRS).

(d) Sum-distance path loss law (Hybrid IRS).

Fig. 5. Achievable rate across the whole area.

opportunistic and nearest association policy, respectively. The parameters for the simulations are:  $\alpha = 4$ ,  $\eta \leq 14$  dB, the total deployment budget is 500, and the deployment

cost of active(passive) reflecting element is 5. Specifically, the number of IRS element is N = 500 for passive IRS, while the number of IRS elements is N = 100 for active



Fig. 6. Performance comparison between nearest and opportunistic association policies across various types of IRS with same power and deployment budget.



Fig. 7. Relationship between the outage probability and rate probability.



Fig. 8. Verification of the approximated channel model for the mixture channel of multiple IRSs-aided scenarios.

IRS [19], [36]. We observed that the performance varies across different configurations, such as association policies, e.g., nearest and opportunistic association, and types of IRS,



i.e., passive/active/hybrid IRS. It is noteworthy that the hybrid IRS offers better performance than passive/active IRS with the same power and deployment budget for both product-/sum-distance path loss laws, under the mentioned setup.

Moreover, there are big performance gaps between the two association policies for all three types of IRSs. These results verifies that even for passive IRS, the nearest IRS might be a sub-optimal option. Although the nearest association provides a similar trend for the passive IRS-aided communication, there is still a big performance gap between these two association policies, which can be explained by the geometric models.

However, for the active and hybrid IRS, these two association policy behaves reversely. If nearest association policy is adopted, the performance of active/hybrid IRS decreases with increasing density of IRS. On the contrary, the performance increases with IRS density when opportunistic association policy is adopted. This is reasonable because the active IRS is supposed to be deployed at a certain distance from the typical UE [21], [48]. When the density of IRS is increasing, the connect distance between the UE and its nearest IRS is decreased, which would degrade the performance enhancement of the active IRS due to the instantaneous amplification and thermal noise introduced by the active parts.

#### D. Outage Probability and Rate Probability

The relationship between outage probability and rate probability, defined as one minus low-rate probability, is illustrated in Fig. 7 for two path loss models. Generally, the curves for the sum-distance model are more sensitive to changes in  $\frac{\tau}{e^{r_0}-1}$ , which is an essential factor in (33). The curves fall into three distinct categories: a) the outage probability and rate probability are equal given  $\frac{\tau}{e^{r_0}-1} = 1$ , b) the outage probability exceeds the rate probability when  $\frac{\tau}{e^{r_0}-1} < 1$ , and c) the outage probability is lower than the rate probability for the same rate probability increases with  $\frac{\tau}{e^{r_0}-1}$ . The effect of path loss exponents is presented for both sum- and product-distance path loss laws. The results indicate that as the path loss exponent increases, the benefit of IRSs is limited due to severe path loss of the cascaded channels.

## *E. Channel Modeling of the Mixture Channel of Multiple IRSs*

In Fig. 8, we present the distribution for small-scale channel coefficients of the mixture channel composed by multiple serving IRSs. It is observed that the approximated channel distribution proposed in Lemma 4 closely aligns with the empirical results. Furthermore, as L increases, the PDF of the channel coefficient exhibits reduced kurtosis, and the mixture channel becomes stronger and provides a greater channel gain. Thus, the utilization of multiple distributed IRSs in communication enables greater flexibility in deploying the IRSs, resulting in comparable performance to that of a single centralized serving IRS with the same total reflecting elements.

#### VII. CONCLUSION

In this work, we studied the geometric models to characterize the cascaded channel's integrated path loss distance in IRS-aided communications. We derived the statistical distribution of the integrated path loss distance and applied these geometric models to determine the hybrid IRS's optimal locations, which encompass both passive and active IRS as special cases. Our findings indicate that the hybrid IRS's optimal placement depends on the integrated path loss distance of the cascaded channel and the network parameters. This solution can be utilized to develop an opportunistic association policy for the IRS and strategic deployment planning. The geometric models offer a new perspective for analyzing IRS-aided networks' performance, allowing accurate statistical modeling of the integrated cascaded channel. Furthermore, we studied several scenarios to explore the potential implementations of the geometric model.

## APPENDIX A

In this appendix, we provide a proof of Theorem 1. The area calculation of the Cassini oval is divided into two cases:  $e_{\rm P} \ge 1$  and  $e_{\rm P} < 1$ , as shown below, where  $e_{\rm P} = \frac{\sqrt{d_{\rm BIU,P}}}{c}$ .

Case 1: When  $e_P \ge 1$ , the curve is continuous, and the polar radius of a Cassini Oval is

$$r^{2} = c^{2} \left[ \cos(2\theta) + \sqrt{e_{\rm P}^{4} - \sin^{2}(2\theta)} \right],$$
 (46)

and then, the area is given by

$$\begin{split} S &= 4 \int_0^{\frac{\pi}{2}} \frac{1}{2} r^2 \mathrm{d}\theta \\ &= 2c^2 \left[ \int_0^{\frac{\pi}{2}} \cos(2\theta) \mathrm{d}\theta + \int_0^{\frac{\pi}{2}} \sqrt{\mathrm{e}_{\mathrm{P}}^4 - \sin^2(2\theta)} \mathrm{d}\theta \right] \\ &= 2c^2 \left[ \int_0^{\frac{\pi}{4}} \sqrt{\mathrm{e}_{\mathrm{P}}^4 - \sin^2(2\theta)} \mathrm{d}\theta + \int_{\frac{\pi}{4}}^{\frac{\pi}{2}} \sqrt{\mathrm{e}_{\mathrm{P}}^4 - \sin^2(2\theta)} \mathrm{d}\theta \right] \\ &= 2c^2 \int_0^{\frac{\pi}{2}} \mathrm{e}_{\mathrm{P}}^2 \sqrt{1 - \mathrm{e}_{\mathrm{P}}^{-2} \sin^2(\theta)} \mathrm{d}\theta = 2d_{\mathrm{BIU},\mathrm{P}} E(\mathrm{e}_{\mathrm{P}}^{-2}), \end{split}$$

where  $E(k) = \int_0^{\frac{\pi}{2}} \sqrt{1 - k^2 \sin^2 \theta} d\theta$  is the complete elliptic integral of the second kind [39].

Case 2: When  $e_P < 1$ , the curve is discontinuous, and the polar radius of a Cassini Oval is given by

$$r_{1,2}^2 = c^2 \left[ \cos(2\theta) \pm \sqrt{\mathrm{e}_{\mathrm{P}}^4 - \sin^2(2\theta)} \right].$$
 (47)

The range of  $\theta$  is given by  $e_{\rm P}^4 = \sin^2(2\theta)$ , which results in  $\theta \in [-\theta_0, \theta_0]$  and  $\theta_0 = \frac{1}{2} \arcsin(e_{\rm P}^2)$ . Then, the area is

$$S = 4 \int_0^{\theta_0} \frac{1}{2} (r_1^2 - r_2^2) \mathrm{d}\theta, \tag{48}$$

which can be further derived as

$$\begin{split} S &= 4c^2 \int_0^{\theta_0} \sqrt{\mathbf{e}_{\mathbf{P}}^4 - \sin^2(2\theta)} \mathrm{d}\theta \\ &= 2c^2 \int_0^{2\theta_0} \sqrt{\mathbf{e}_{\mathbf{P}}^4 - \sin^2(\theta)} \mathrm{d}\theta \\ &= 2d_{\mathrm{BIU,P}} E\left[\sin^{-1}(\mathbf{e}_{\mathbf{P}}^2), \mathbf{e}_{\mathbf{P}}^{-2}\right] \\ \stackrel{(a)}{=} 2d_{\mathrm{BIU,P}} \int_0^1 \frac{1 - x^2}{\sqrt{1 - \mathbf{e}_{\mathbf{P}}^4 x^2} \sqrt{1 - x^2}} \mathbf{e}_{\mathbf{P}}^2 \mathrm{d}x \\ &= 2c^2 \mathbf{e}_{\mathbf{P}}^4 \left[ K(\mathbf{e}_{\mathbf{P}}^2) - \int_0^1 \frac{x^2}{\sqrt{1 - \mathbf{e}_{\mathbf{P}}^4 x^2} \sqrt{1 - x^2}} \mathrm{d}x \right] \\ &= 2c^2 \left[ E(\mathbf{e}_{\mathbf{P}}^2) - (1 - \mathbf{e}_{\mathbf{P}}^4) K(\mathbf{e}_{\mathbf{P}}^2) \right], \end{split}$$

where step (a) follows a substitution of  $x = e_{\rm P}^{-2}t$ ,  $E(\phi, k) = \int_0^{\phi} \sqrt{1 - k^2 \sin^2 \theta} d\theta$  is the incomplete elliptic integral of the second kind, and  $K(k) = \int_0^{\frac{\pi}{2}} \frac{1}{\sqrt{1 - k^2 \sin^2 \theta}} d\theta$  is the complete elliptic integral of the first kind [39]. With some mathematical simplification, the results in (4) is achieved. This completes the proof.

## APPENDIX B

In this appendix, we provide a proof of Lemma 2. The CDF can be derived using the area ratio, and then PDF is achieved by taking derivative of the CDF with the aid of  $\frac{d}{dk}E(k) = \frac{E(k)-K(k)}{k}$ , and  $\frac{d}{dk}K(k) = \frac{E(k)}{k(1-k^2)} - \frac{K(k)}{k}$  [39]. This completes the proof.

#### APPENDIX C

In this appendix, we provide a proof of Lemma 3. For product-distance path loss law, given the  $d_{\text{BIU}}$ , the coordinates of the optimal location of the hybrid passive and active IRS satisfies

$$(x^{2} + y^{2})^{2} - 2c^{2}(x^{2} - y^{2}) = d_{BIU}^{2} - c^{4},$$
(49)

$$(x+c)^2 + y^2 = \left(\frac{P_{\rm T}\delta^2}{P_{\rm F}\delta_{\rm F}^2}\right)^{\frac{1}{\alpha}} d_{\rm BIU}.$$
 (50)

Thus, we can achieve that

$$y^{2} = \left(\frac{P_{\rm T}\delta^{2}}{P_{\rm F}\delta_{\rm F}^{2}}\right)^{\frac{1}{\alpha}} d_{\rm BIU} - (x+c)^{2}.$$
 (51)

Next, substitute (51) into (49) and (50), (23) can be achieved with some mathematical simplifications. Following similar procedure, the results for sum-distance path loss law can be achieved. This completes the proof.

#### APPENDIX D

In this appendix, we provide a proof for Proposition 2. As the channel of each individual link, e.g., BS $\rightarrow$ IRS or IRS $\rightarrow$ UE, follows generalized fading distribution, the channel gain of the cascaded channel can be modeled as a mixture Gamma distribution as proved in [9]. Then, the PDF of mixture Nakagami-*m* distribution for channel coefficient of the cascaded link (BS $\rightarrow$ IRS $\rightarrow$ UE) in (37) can be achieved with the assistance of the PDF scaling law  $f_X(x) = 2xf_S(x^2)$ , where  $x = \sqrt{s}$ . This completes the proof.

#### APPENDIX E

In this appendix, we provide a proof for the approximated channel modeling for the mixture channel of L IRSs aided mixture channel in Lemma 4. As the channel coefficient of each cascaded link follows mixture Nakagami-m distribution, the average channel coefficient of the mixture Nakagami-m distributed link is derived as follows:

$$\mathbb{E}[X] = \sum_{i=1}^{I} \varepsilon_i \int_0^\infty x^{2\beta_i - 1} e^{-\xi_i x^2} \cdot 2x \mathrm{d}x$$
$$\stackrel{(a)}{=} \sum_{i=1}^{I} \varepsilon_i \int_0^\infty t^{\beta_i - \frac{1}{2}} e^{-\xi_i t} \mathrm{d}t \stackrel{(b)}{=} \sum_{i=1}^{I} \varepsilon_i \left(\beta_i - \frac{1}{2}\right)! \xi_i^{-\beta_i - \frac{1}{2}},$$

where step (a) is achieved by a substitution of  $t = x^2$ , and step (b) is derived with the assistance of [39, eq.3.351.5]. As phase alignment is employed, the mean of the channel coefficient is

$$\mu = \mathbb{E}[|h_{\rm S}|] = \mathbb{E}[|h_{\rm BIU,1}|] + \dots + \mathbb{E}[|h_{\rm BIU,L}|] = \sum_{l=1}^{L} \mu_l,$$
(52)

where  $\mu_l = \mathbb{E}[|h_{\text{BIU},l}|]$ , and its variance is given by

$$\delta_{\rm S}^2 = \mathbb{E}[|h_{\rm S}|^2] - \mu^2, \tag{53}$$

with

$$\mathbb{E}[|h_{\rm S}|^2] = \mathbb{E}\left[\sum_{l_1=1}^L \sum_{l_2=1}^L h_{l_1} h_{l_2}\right] = L + \sum_{l_1=1}^L \sum_{l_2=1, l_2 \neq l_1}^L \mu_{l_1} \mu_{l_2},$$

where  $h_{l_1}$  and  $h_{l_2}$  represent the channel of  $l_1$ -th and  $l_2$ -th BS $\rightarrow$ IRS $\rightarrow$ UE link, respectively. This completes the proof.

#### REFERENCES

- Q. Wu et al., "Intelligent reflecting surface-aided wireless communications: A tutorial," *IEEE Trans. Commun.*, vol. 69, no. 5, pp. 3313–3351, May 2021.
- [2] M. Di Renzo et al., "Smart radio environments empowered by reconfigurable intelligent surfaces: How it works, state of research, and the road ahead," *IEEE J. Sel. Areas Commun.*, vol. 38, no. 11, pp. 2450–2525, Nov. 2020.
- [3] W. Qingqing and Z. Rui, "Towards smart and reconfigurable environment: Intelligent reflecting surface aided wireless network," *IEEE Commun. Mag.*, vol. 58, no. 1, pp. 106–112, Jan. 2019.
- [4] S. Gong et al., "Toward smart wireless communications via intelligent reflecting surfaces: A contemporary survey," *IEEE Commun. Surveys Tuts.*, vol. 22, no. 4, pp. 2283–2314, 4th Quart., 2020.
- [5] M. A. ElMossallamy, H. Zhang, L. Song, K. G. Seddik, Z. Han, and G. Y. Li, "Reconfigurable intelligent surfaces for wireless communications: Principles, challenges, and opportunities," *IEEE Trans. Cognit. Commun. Netw.*, vol. 6, no. 3, pp. 990–1002, Sep. 2020.
- [6] J. Lyu and R. Zhang, "Spatial throughput characterization for intelligent reflecting surface aided multiuser system," *IEEE Wireless Commun. Lett.*, vol. 9, no. 6, pp. 834–838, Jun. 2020.
- [7] J. Lyu and R. Zhang, "Hybrid active/passive wireless network aided by intelligent reflecting surface: System modeling and performance analysis," *IEEE Trans. Wireless Commun.*, vol. 20, no. 11, pp. 7196–7212, Nov. 2021.
- [8] Q. Wu and R. Zhang, "Intelligent reflecting surface enhanced wireless network via joint active and passive beamforming," *IEEE Trans. Wireless Commun.*, vol. 18, no. 11, pp. 5394–5409, Nov. 2019.
- [9] Y. Li and Y. J. Chun, "Analysis of IRS-assisted downlink wireless networks over generalized fading," *IEEE Trans. Wireless Commun.*, early access, Mar. 4, 2024, doi: 10.1109/TWC.2024.3369662.
- [10] R. Long, Y.-C. Liang, Y. Pei, and E. G. Larsson, "Active reconfigurable intelligent surface aided wireless communications," *IEEE Trans. Wireless Commun.*, vol. 20, no. 8, pp. 4962–4975, Aug. 2021.
- [11] Z. Zhang et al., "Active RIS vs. passive RIS: Which will prevail in 6G?" *IEEE Trans. Commun.*, vol. 71, no. 3, pp. 1707–1725, Mar. 2023.
- [12] K. Zhi, C. Pan, H. Ren, K. K. Chai, and M. Elkashlan, "Active RIS versus passive RIS: Which is superior with the same power budget?" *IEEE Commun. Lett.*, vol. 26, no. 5, pp. 1150–1154, May 2022.
- [13] K. Liu, Z. Zhang, L. Dai, S. Xu, and F. Yang, "Active reconfigurable intelligent surface: Fully-connected or sub-connected?" *IEEE Commun. Lett.*, vol. 26, no. 1, pp. 167–171, Jan. 2022.
- [14] M. H. Khoshafa, T. M. N. Ngatched, M. H. Ahmed, and A. R. Ndjiongue, "Active reconfigurable intelligent surfaces-aided wireless communication system," *IEEE Commun. Lett.*, vol. 25, no. 11, pp. 3699–3703, Nov. 2021.
- [15] G. Chen, Q. Wu, C. He, W. Chen, J. Tang, and S. Jin, "Active IRS aided multiple access for energy-constrained IoT systems," *IEEE Trans. Wireless Commun.*, vol. 22, no. 3, pp. 1677–1694, Mar. 2023.
- [16] J. Lonč ar, A. Grbic, and S. Hrabar, "Ultrathin active polarizationselective metasurface at X-band frequencies," *Phys. Rev. B, Condens. Matter*, vol. 100, no. 7, Aug. 2019, Art. no. 075131.
- [17] J. Bousquet, S. Magierowski, and G. G. Messier, "A 4-GHz active scatterer in 130-nm CMOS for phase sweep amplify-and-forward," *IEEE Trans. Circuits Syst. I, Reg. Papers*, vol. 59, no. 3, pp. 529–540, Mar. 2012.
- [18] K. K. Kishor and S. V. Hum, "An amplifying reconfigurable reflectarray antenna," *IEEE Trans. Antennas Propag.*, vol. 60, no. 1, pp. 197–205, Jan. 2012.
- [19] Z. Kang, C. You, and R. Zhang, "Active-passive IRS aided wireless communication: New hybrid architecture and elements allocation optimization," *IEEE Trans. Wireless Commun.*, early access, 2024, doi: 10.1109/TWC.2023.3308373.
- [20] R. Schroeder, J. He, G. Brante, and M. Juntti, "Two-stage channel estimation for hybrid RIS assisted MIMO systems," *IEEE Trans. Commun.*, vol. 70, no. 7, pp. 4793–4806, Jul. 2022.
- [21] C. You and R. Zhang, "Wireless communication aided by intelligent reflecting surface: Active or passive?" *IEEE Wireless Commun. Lett.*, vol. 10, no. 12, pp. 2659–2663, Dec. 2021.

- [22] J. Huang et al., "Reconfigurable intelligent surfaces: Channel characterization and modeling," *Proc. IEEE*, vol. 110, no. 9, pp. 1290–1311, Sep. 2022.
- [23] M. Al-Jarrah, A. Al-Dweik, E. Alsusa, Y. Iraqi, and M.-S. Alouini, "On the performance of IRS-assisted multi-layer UAV communications with imperfect phase compensation," *IEEE Trans. Commun.*, vol. 69, no. 12, pp. 8551–8568, Dec. 2021.
- [24] M. Elhattab, M. A. Arfaoui, C. Assi, and A. Ghrayeb, "RIS-assisted joint transmission in a two-cell downlink NOMA cellular system," *IEEE J. Sel. Areas Commun.*, vol. 40, no. 4, pp. 1270–1286, Apr. 2022.
- [25] M. Alayasra and H. Arslan, "IRS-enabled beam-space channel," *IEEE Trans. Wireless Commun.*, vol. 21, no. 6, pp. 3822–3835, Jun. 2022.
- [26] X. Liu, E. K. P. Chong, and N. B. Shroff, "A framework for opportunistic scheduling in wireless networks," *Comput. Netw.*, vol. 41, no. 4, pp. 451–474. 2003.
- [27] D. Tse and P. Viswanath, *Fundamentals of Wireless Communication*. Cambridge, U.K.: Cambridge Univ. Press, 2005.
- [28] J. G. Andrews, A. K. Gupta, and H. S. Dhillon, "A primer on cellular network analysis using stochastic geometry," 2016, arXiv:1604.03183.
- [29] J. G. Andrews et al., "A tractable approach to coverage and rate in cellular networks," *IEEE Trans. Commun.*, vol. 59, no. 11, pp. 3122–3134, Nov. 2011.
- [30] M. Haenggi, Stochastic Geometry for Wireless Networks. Cambridge, U.K.: Cambridge Univ. Press, 2012.
- [31] C. You, B. Zheng, W. Mei, and R. Zhang, "How to deploy intelligent reflecting surfaces in wireless network: BS-side, user-side, or both sides?" J. Commun. Inf. Netw., vol. 7, no. 1, pp. 1–10, 2022.
- [32] W. Tang et al., "Wireless communications with reconfigurable intelligent surface: Path loss modeling and experimental measurement," *IEEE Trans. Wireless Commun.*, vol. 20, no. 1, pp. 421–439, Jan. 2021.
- [33] Y. Fang, S. Atapattu, H. Inaltekin, and J. Evans, "Optimum reconfigurable intelligent surface selection for wireless networks," *IEEE Trans. Commun.*, vol. 70, no. 9, pp. 6241–6258, Sep. 2022.
- [34] Q. Wang, B. Rengarajan, and J. Widmer, "Increasing opportunistic gain in small cells through energy-aware user cooperation," *IEEE Trans. Wireless Commun.*, vol. 13, no. 11, pp. 6356–6369, Nov. 2014.
- [35] J. Lončar and Z. Šipuš, "Challenges in design of power-amplifying active metasurfaces," in *Proc. Int. Symp. (ELMAR)*, Zadar, Croatia, Oct. 2020, pp. 9–12.
- [36] F. Amato, C. W. Peterson, B. P. Degnan, and G. D. Durgin, "Tunneling RFID tags for long-range and low-power microwave applications," *IEEE J. Radio Freq. Identificat.*, vol. 2, no. 2, pp. 93–103, Jun. 2018.
- [37] S. Atapattu, C. Tellambura, and H. Jiang, "A mixture gamma distribution to model the SNR of wireless channels," *IEEE Trans. Wireless Commun.*, vol. 10, no. 12, pp. 4193–4203, Dec. 2011.
- [38] R. A. DeVore and G. G. Lorentz, *Constructive Approximation*. Berlin, Germany: Spring-Verlag, 1993.
- [39] M. Abramowitz and I. A. Stegun, Eds., Handbook of Mathematical Functions With Formulas, Graphs, and Mathematical Tables, vol. 55. U.S. Government Printing Office, 1968.
- [40] A. L. Moustakas, G. C. Alexandropoulos, and M. Debbah, "Reconfigurable intelligent surfaces and capacity optimization: A large system analysis," *IEEE Trans. Wireless Commun.*, early access, Apr. 13, 2023, doi: 10.1109/TWC.2023.3265420.
- [41] Y. Ren et al., "On deployment position of RIS in wireless communication systems: Analysis and experimental results," *IEEE Wireless Commun. Lett.*, vol. 12, no. 10, pp. 1756–1760, Oct. 2023, doi: 10.1109/LWC.2023.3292125.
- [42] M. Berger, Geometry I. Springer, 2009.

- [43] S. Zhang and R. Zhang, "Intelligent reflecting surface aided multi-user communication: Capacity region and deployment strategy," *IEEE Trans. Commun.*, vol. 69, no. 9, pp. 5790–5806, Sep. 2021.
- [44] W. He, D. He, X. Ma, X. Chen, Y. Fang, and W. Zhang, "Joint user association, resource allocation, and beamforming in RIS-assisted multiserver MEC systems," *IEEE Trans. Wireless Commun.*, early access, 2023.
- [45] M. Xia and S. Aissa, "Cooperative AF relaying in spectrum-sharing systems: Performance analysis under average interference power constraints and Nakagami-*m* fading," *IEEE Trans. Commun.*, vol. 60, no. 6, pp. 1523–1533, Jun. 2012.
- [46] Y. J. Chun, S. L. Cotton, H. S. Dhillon, F. J. Lopez-Martinez, J. F. Paris, and S. K. Yoo, "A comprehensive analysis of 5G heterogeneous cellular systems operating over κ-μ shadowed fading channels," *IEEE Trans. Wireless Commun.*, vol. 16, no. 11, pp. 6995–7010, Nov. 2017.
- [47] K. A. Hamdi, "A useful technique for interference analysis in Nakagami fading," *IEEE Trans. Commun.*, vol. 55, no. 6, pp. 1120–1124, Jun. 2007.
- [48] Y. Li, C. You, and Y. J. Chun, "Active-IRS aided wireless network: System modeling and performance analysis," *IEEE Commun. Lett.*, vol. 27, no. 2, pp. 487–491, Feb. 2023.



Yunli Li received the B.S. degree in electronic information engineering from Nanjing University of Aeronautics and Astronautics in 2018 and the M.S. degree in telecommunications from The Hong Kong University of Science and Technology in 2019. She is currently pursuing the Ph.D. degree with the Department of Electronic Engineering, City University of Hong Kong. Her research interests include intelligent reflecting surfaces, channel modeling, and stochastic geometry.



Young Jin Chun (Member, IEEE) received the B.S. degree in electrical engineering from Yonsei University in 2004, the M.S. degree in electrical engineering from the University of Michigan in 2007, and the Ph.D. degree in electrical engineering from Iowa State University in 2011. From September 2011 to October 2018, he was with Sungkyunkwan University, Qatar University, and Queen's University Belfast in various academic positions. He joined the Department of Electronic Engineering, City University of Hong Kong, in November 2018, as an

Assistant Professor of wireless communication. His primary research interests include the area of wireless communications with an emphasis on stochastic geometry, system-level network analysis, device-to-device communications, and various use cases of 5G communications.